Author's response to reviews

Title: Rationale And Design Of The Plate Or Pin (POP) Study For Dislocated Midshaft Clavicular Fractures: study protocol for a randomised controlled trial

Authors:

Frans J.G. Wijdicks (fjwijdicks@gmail.com)
Marijn R. Houwert (marijnhouwert@hotmail.com)
Marcel G.W. Dijkgraaf (m.g.dijkgraaf@amc.uva.nl)
Diederik H. De Lange (ddelange@diakhuis.nl)
Sven A.G. Meylaerts (s.meylaerts@mchaaglanden.nl)
Michiel H.J. Verhofstad (m.verhofstad@elisabeth.nl)
Egbert J.J.M. Verleisdonk (ejverleisdonk@diakhuis.nl)

Version: 6 Date: 23 June 2011

Author's response to reviews: see over
Dear Editors,

Hereby we would like to submit our final revision of the manuscript entitled: “Rationale en design of the Plate Or Pin (POP) study for dislocated midshaft clavicular fractures: study protocol for a randomised controlled trial”. We have adjusted the manuscript accordingly to the reviewers comments.

First of all we would like to thank the reviewers again for the time and effort they took for reviewing our manuscript.

Please remove the job titles/authors' information from the auth list. If necessary, this information can be given in a section entitled 'Author's Information' after the authors' contributions.

The job titles/authors' information has been removed.

Tables: We notice you have included a list as a table, (table 2) please reformat so it is incorporated within the main text. Please remember to update any citations to it in the text. Alternatively format into a table with two or more columns.

Table is formatted into two columns

References: the reference list should contain up to 30 authors names for each citation. The term 'et al.' should not be used if there are fewer than 30 authors.

Reference list is adapted and every author up to 30 authors has been added.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Manuscript is checked for errors.

All authors agreed with the changes that were made. The authors hope these revisions are to your approval and the authors look forward hearing from you.

Yours sincerely, on behalf of all authors,

Frans-Jasper Wijdicks

F.J.G. Wijdicks
Griftstraat 34
3572 GW Utrecht, The Netherlands
Mob: +31651957012
Email: fjgwijdicks@gmail.com